Author's response to reviews

Title: Applying cusum-based methods for the detection of outbreaks of Ross River virus disease in Western Australia

Authors:

Rochelle E Watkins (Rochelle.Watkins@curtin.edu.au)
Serryn Eagleson (S.Eagleson@curtin.edu.au)
Bert Veenendaal (B.Veenendaal@curtin.edu.au)
Graeme Wright (G.L.Wright@curtin.edu.au)
Aileen J Plant (Rochelle.Watkins@curtin.edu.au)

Version: 3 Date: 21 July 2008

Author's response to reviews:

Dear Dr Todd,
Thank you very much for your email. I have revised the manuscript text on page 5 para 3 to indicate that the data are not publicly available, and that de-identified data were obtained under agreement with the relevant departments of health. A reference to ethical approval of the study has also been added on page 10 para 2.
I have checked the manuscript formatting against the checklist.
Thank you again.
Regards,
Rochelle